Author’s response to reviews

Title: Challenges in the provision of kidney care at the largest public nephrology center in Guatemala: A qualitative study with health professionals

Authors:

David Flood (david@wuqukawoq.org)
Katharine Wilcox (katie@wuqukawoq.org)
Andrea Aguilar Ferro (andreaixtoloc@gmail.com)
Carlos Mendoza Montano (cmendoza@incap.int)
Joaquin Barnoya (barnoyaj@gmail.com)
Pablo Garcia (pablo.r.garcia.hu@gmail.com)
Randall Lou-Meda (randall_lou@yahoo.com)
Peter Rohloff (peter@wuqukawoq.org)
Anita Chary (achary@partners.org)

Version: 2 Date: 05 Feb 2020

Author’s response to reviews:

Assistant Editor Comments:

1. Corresponding author email

We note that the corresponding authors email on the title page of your manuscript (anita@wuqukawoq.org) is different to the one entered into the submission system (achary@partners.org). Please amend one of these entries and ensure that the corresponding authors email entered into the submission system and into the title page of your manuscript are the same. Please only list one email on the title page of your manuscript.

--This has been changed to “achary@partners.org”

2. Consent to Participate

In your “Ethical approval and consent to participate” section, please confirm whether informed consent to participate, written or verbal, was obtained from all of the participants in the study and clearly state this in your manuscript. Please note that in the case of minors, which refers to individuals younger than the age of 16, consent to participate must be obtained from their parents or legal guardians. As such, we ask that in your “Ethics approval and consent to participate” section, to clarify whether informed consent to participate,
written or verbal, was obtained from the parents or legal guardians of any participant under the age of 16 and clearly state this in your manuscript. If consent was verbal, please state the reason and whether the ethics committee approved this procedure. If the need for consent was waived by an Institutional Review Board (IRB) or is deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation.

--We have added text to clarify that all participants were above the age of 18, the reason for which written/signed consent forms were not collected, and that the verbal consent process was approved by local IRBs.

3. Consent for publication

Currently, the statement in your “Consent for publication” section of your declarations is incorrect. Please note that, for all manuscripts that include details, images, or videos relating to an individual person, written informed consent for the publication of these details must be obtained from that person. As such, we ask that in your “Consent for publication” section, to please clarify whether written informed consent for publication of clinical details and/or clinical images was obtained from the all of the participants.

Please note that in the case of minors, which refers to individuals younger than the age of 18, consent for publication must be obtained from their parents or legal guardians. As such, we ask that in your “Consent for publication” section, to please clarify whether written informed consent for publication of clinical details and/or clinical images was obtained from the parents or legal guardians of any participant under the age of 18.

--We have changed this section to convey that our manuscript does not include clinical details, images, or videos relating to an individual person. We also convey that study participants agreed to publication of interview quotations through the verbal consent process.

4. Authors’ contributions

a) We have noticed a discrepancy in the Authors’ contributions section of your manuscript. In regard to the individual authors’ contributions, you have not stated the individual contribution of Pablo Garcia. Please include the individual contribution of Pablo Garcia in the Authors’ contributions section of your manuscript. Please also represent author’s names using their full initials, not their name, and without any titles. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors. Guidance and criteria for authorship can be found here: http://www.biomedcentral.com/submissions/editorial-policies#authorship

--We have added in Pablo Garcia’s contributions.

b) Please amend the initials “AA” to “AAF”.

--We have amended the initials.
5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Should you wish to respond to these revision requests, please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter.

--This has been done.